Author’s response to reviews

Title: The psychometric properties of the Barkin index of maternal functioning (BIMF) for the Iranian population

Authors:
Mojgan Mirghafourvand (mirghafourvand@gmail.com)
Jennifer Barkin (barkin jl@mercer.edu)
Mohammad Jafarabadi (m.asghari862@gmail.com)
Fatemeh Karami (fatemeh.karami1995@gmail.com)
Solmaz Ghanbari-Homayi (homayisolmaz@gmail.com)

Version: 2 Date: 27 Nov 2019

Author’s response to reviews:

Dear Editor of BMC Women's Health,

Please find the revised manuscript entitled, “The psychometric properties of the Barkin index of maternal functioning (BIMF) for the Iranian population”. We have carefully considered the comments and have addressed them below.

Yours sincerely,

Solmaz Ghanbari-Homayi, Midwifery Department, Tabriz University of Medical Sciences, Tabriz, Iran. Postal address: 5138947977, Fax number: 041-34796969, Email: homayisolmaz@gmail.com

Editor Comments:

1.Funding

- In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

The role of the funding body have been added.
2. Clean copy

- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.

A single, final, clean version have been uploaded.

Thank you so much